Author’s response to reviews

Title: Incorporating Clinical Guidelines Through Clinician Decision Making

Authors:

Paul R. Falzer Ph.D. (paul.falzer@yale.edu)
Brent A. Moore Ph.D. (brent.moore@yale.edu)
D. Melissa Garman MSW, DCSW (melissa.garman@po.state.ct.us)

Version: 2 Date: 11 January 2008

Author’s response to reviews:

1. Revisions of the abstract and text, as recommended by Dr. Robbie Foy on 1/11/08.

2. Attachment of latest approval from the Veterans Administration HSS, 10/4/07.

3. Attachment of the notice of award from NIH, obtained from CRISP. NIH does not give a formal notice of award. If the attachment provided isn't sufficient, I will obtain a transmittal letter (which pertains to transfer of funds) from our Grants & Contracts office. The CRISP document is publicly available at the following Website (remove spaces and paste the entire URL into a browser):
http://crisp.cit.nih.gov/crisp/CRISP_LIB.getdoc?
textkey=7237844&p_grant_num=5R34MH070871-03&p_query=&ticket=54143789&p_audit_session_id=285728544&p_keywords=